**Author's response to reviews**

**Title:** Statistical Validation of the Criteria for Symptom Remission in Schizophrenia: Preliminary Findings.

**Authors:**

Mark G. A. Opler (mark.opler@panss.org)
Lawrence Yang (lyang@panss.org)
Sue Caleo (SCALEO1@PSMBE.JNJ.COM)
Philip Alberti (philipalberti@gmail.com)

**Version:** 5  **Date:** 29 June 2007

**Author's response to reviews:**

To: BMC Psychiatry

From: Mark Opler

Re: Manuscript

Dear Editors,

The newly revised manuscript, entitled "Statistical Validation of the Criteria for Symptom Remission in Schizophrenia: Preliminary Findings" incorporates the comments and guidance of your editorial staff.

1. The word "Finally" has been removed from the last paragraph.
2. Grammar and spelling errors have been corrected throughout the manuscript.
3. We have reviewed the manuscript checklist.

We are grateful for the opportunity to publish in BMC Psychiatry and thank you for your patience and assistance during the peer review process.

Sincerely,

-Mark